Author's response to reviews

Title: Assessment of lower urinary symptom flare with overactive bladder symptom score and International Prostate Symptom Score in patients treated with iodine-125 implant brachytherapy: Long-term follow-up experience at a single institute

Authors:
Makito Miyake (makitomiyake@yahoo.co.jp)
Nobumichi Tanaka (sendo@naramed-u.ac.jp)
Isao Asakawa (iasakawa@naramed-u.ac.jp)
Shunta Hori (horimaus@gmail.com)
Yosuke Morizawa (tigers.yosuke@gmail.com)
Yoshihiro Tatsumi (takuro.birds.nest@gmail.com)
Yasushi Nakai (nakaiyasusiuro@live.jp)
Takeshi Inoue (you1513tt@yahoo.co.jp)
Satoshi Anai (sanai@naramed-u.ac.jp)
Kazumasa Torimoto (torimoto@nmu-gw.naramed-u.ac.jp)
Katsuya Aoki (aokik@nmu-gw.naramed-u.ac.jp)
Masatoshi Hasegawa (hasegawa@naramed-u.ac.jp)
Tomomi Fujii (fuijti@naramed-u.ac.jp)
Noboru Konishi (nkonishi@naramed-u.ac.jp)
Kiyohide Fujimoto (kiyokun@naramed-u.ac.jp)

Version: 2 Date: 25 May 2017

Author’s response to reviews:

We appreciate your considerable editorial work. We have decided to submit the revised manuscript to BMC Urology. Here is response to the comments.

1. Please provide clarification on whether the informed consent was verbal or written.
If verbal, please provide justification for why verbal consent was obtained and not written.

>

In the Declaration, we have added “written informed consent” in the section of Ethics approval and consent to participate.

2. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

>

All authors have been involved in drafting the manuscript and revising it critically for important intellectual content and approved the version to be published. In the end of text, we have modified the authors’ contribution in more detail in the section of Authors’ contributions.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

>

We appreciate your editorial work. We have completed the files and resubmitted to your journal.